Author’s response to reviews

Title: The role of a specialist neurological infection service at a university teaching hospital in Northwest England

Authors:

Lance Turtle (lturtle@liverpool.ac.uk)
Agam Jung (Agam.Jung@leedsth.nhs.uk)
Nicholas J Beeching (Nicholas.Beeching@rlbuht.nhs.uk)
Derek Cocker (d.cocker@doctors.org.uk)
Gerry R Davies (gerrydavies@doctors.org.uk)
Andrew Nicolson (andrew.nicolson@thewaltoncentre.nhs.uk)
Michael BJ Beadsworth (Mike.Beadsworth@rlbuht.nhs.uk)
Alastair RO Miller (alastairmiller@me.com)
Tom Solomon (tsolomon@liv.ac.uk)

Version: 2  Date: 6 March 2015

Author’s response to reviews:

Dear BMC editorial staff,

In response to initial assessment by editorial staff the following changes have been made to the manuscript:

1). Line numbering added.
2). Title capitalisation checked to ensure compliance with BMC style
3). Page breaks removed.
4). Title page updated to include email addresses (full postal addresses are already present).

The text of the manuscript is otherwise unchanged.

Yours sincerely,

Dr Lance Turtle